\documentclass[12pt]{article}
\usepackage{hpshsetup_E}
%------------------------------------------------------
%-------------- Insert These's Title Name---------------
\newcommand{\TitleName}{We are all good citizens.}

\begin{document}

\section{Introduction}
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen \autocite{dirac}.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.


\section{Literature Review}

I am a good citizen. I am a good citizen. I am a good citizen.
See Table~\ref{tbl:pen},
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.

\bxtable[htp]{Stationery comsume in HPSH in January\label{tbl:pen}}
{
\begin{tabular}{clcrr}
\toprule
Item & Name & Quantities & Unit price & Amount \\
\midrule
1 & pencil  & 12 & 2.0 & 24.0 \\
2 & fountain pen & 3 & 100.0& 300.0 \\
3 & glue & 5 & 10.0 & 50.0 \\
4 & white paper & 30 & 100 & 3000.0\\
\bottomrule
\end{tabular}}
{Sources: \fullcite{knuth-fa}}

\subsection{First subsection}
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen.
I am a good citizen \autocite{knuth-fa}.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
\textcite{einstein} said:\textbf{``
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.''}
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.

\subsection{Second subsection}

I am a good citizen. I am a good citizen. I am a good citizen.
See Figure~\ref{fig:rope},
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.


 
\bxfigure[htp]{
Wave propagation on elastic string.
\label{fig:rope}}
{\includegraphics[width=.7\linewidth]{wave_inhomo_0276}}
{Sources: \fullcite{unsplash}}


\section{Research Methods}
I am a good citizen. I am a good citizen. I am a good citizen.
See Figure~\ref{fig:film},
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.

\bxfigure[htp]{Old films.\label{fig:film}}
{
\includegraphics[width=.45\linewidth]{50603.jpg}
\includegraphics[width=.45\linewidth]{50608.jpg}
}
{Sources: \fullcite{knuthwebsite}}

I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
\begin{equation}
x = \dfrac{-b \pm \sqrt{b^2-4ac}}{2a}
\label{eqn:formula}
\end{equation}
I am a good citizen. I am a good citizen. I am a good citizen.
Eq.~\ref{eqn:formula} can quickly solve quadratic equation.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.




\bxfigure[htp]{Interference pattern on water tank.\label{fig:tank}}
{\includegraphics[width=.5\linewidth]{watertank}}
{Sources: \fullcite{fowles}}

I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.

	
\section{Analysis and Results}

I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
\begin{enumerate}
    \item main idea
    \item subject matter
    \item inference
    \item supporting details
\end{enumerate}
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.



\section{Conclusion and Suggestions}

I am a good citizen. I am a good citizen. I am a good citizen.
See Figure~\ref{fig:tank},
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.

\bxfigure[htp]{Baby in mother's arms. \label{fig:kater}}
{\includegraphics[width=.5\linewidth]{katerREDU.jpg}}
{Sources: \fullcite{unsplash}}


\begin{itemize}
\item
I am a good citizen.
\item
You're a good citizen.
\item
They're good citizens.
\item
What's your name?
\end{itemize}

I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen \autocite{ter1971}.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.
I am a good citizen. I am a good citizen. I am a good citizen.


\printbibliography

\end{document}
